Author’s response to reviews

Title: Comparison of lumbar plexus block using the axis in-plane method at the plane of the transverse process and at the articular process: A randomized controlled trial

Authors:

Rui Lv (43721238@qq.com)
Chengcheng Shen (mysupercc@163.com)
Chunyong Yang (29326076@qq.com)
Yan Chen (824868146@qq.com)
Juanjuan Li (1359208194@qq.com)
kaizhi lu (lukaizhi2010@163.com)

Version: 5 Date: 19 Jan 2018

Author’s response to reviews:

Editor Comments:

1. The trial registration should take the following form:

Trial registration: name of registry, trial registration number, registered on date of registration.

Please include the date of registration in the Trial Registration information

Answer: Thanks very much for editor’s comment. It’s a good advice. We have revised it.

2. Ethics approval and consent to participate

It is stated 'Additional file 1: List of the Ethics Committees that approved the study'. However, please remove the ethics approval documents from the file inventory and revise the statement as below:

-Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate in this section

-For Consent to Participate, please state clearly whether written informed consent was obtained from the participants in this section.

Answer: Thanks very much for editor’s comment. It’s a good advice. We have revised it.
3. Consent for publication

Please include a 'Consent for publication' section in Declarations. Please state Not Applicable as your manuscript doesn't contain any individual identifying data, but this section must be present.

Answer: Thanks very much for editor’s comment. It’s a good advice. We have revised it.

4. Availability of data and materials

Please include a 'Availability of data and materials' section in Declarations. Availability of Data and Materials refers to the raw data-set upon which the summary statistics in the manuscript are based.

For your information: Here are some examples of Availability of Data and Materials statements that you may choose to adapt if you wish:

• The datasets generated and analysed during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK TO DATASETS]

• The datasets generated and analysed during the current study are available from the corresponding author on reasonable request.

• The datasets generated and analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC] but are available from the corresponding author on reasonable request.

• The datasets generated and analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC].

Answer: Thanks very much for editor’s comment. It’s a good advice. We have revised it.

5. Funding

Please provide the grant code for the funding if there is any.

Answer: Thanks very much for editor’s comment. There’s no grant code for the funding in our research. This study is supported by departmental funding.

6. Upon resubmission, please make sure that your manuscript contains the following seven sections under the heading 'Declarations' at the end of the main text:

(1) Ethics approval and consent to participate

(2) Consent for publication
(3) Availability of data and material

(4) Competing interests

(5) Funding

(6) Authors' contributions

(7) Acknowledgements

Answer: Thanks very much for editor’s comment. It’s a good advice. We have revised it.

7. Figure 1 and Figure 2

Please state whether Figure 1a and Figure 2a are your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

Should you require an alternative source you may wish to try Wikimedia Commons: http://commons.wikimedia.org/wiki/Main_Page

Answer: Thanks very much for editor’s comment. The Figure 1a and Figure 2a are our own.

8. Figure 3

Please remove the logo (the one with the CONSORT transparent reporting of trials) from the figure as the logo might be under copyright.

Answer: Thanks very much for editor’s comment. It’s a good advice. We have revised it.

9. Table 1 is not referenced in the main text. Please check this and make sure it is properly referenced.

Answer: Thanks very much for editor’s comment. It’s a good advice. We have revised it.

10. Please remove the below files from the file inventory:

Ethics approval and consent to participate.docx

Cover Letter.doc

Answer: Thanks very much for editor’s comment. It’s a good advice. We have revised it.
11. Upon resubmission, Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Please ensure the document is in the final form for publication; please upload only files that are to be published. If you wish to respond to these revision requests, please insert the information in to the designated input box only